Preliminary Call for Pap ers 1

Joint Conference SIGMOD/PODS 2000

Dallas, T exas ˘ M ay 14 - 19, 2000 http://www .seas.smu.edu/sigmod2000

19 th ACM SIGM OD-SIG ACT - SIG ART Sympo sium on Principles of Database System s (PODS 2000) http://www .seas.smu.edu/sigmod2000/podscfp.html

The symposium will focus on new developments in the fundamental aspects of databases. Original research papers on the theory, design, specification, or implementation of databases are solicited. Papers emphasizing new topics or the foundations of emerging areas are especially welcomed.

Important Dates: November 2 Paper titles and short abstracts due.

November 8 Papers due.

February 8 Notification about acceptance/rejection.

March 5 Camera-ready due.

2000 ACM SIGM OD International Conference on M anagement of Data (SIGM OD 2000) http://www .seas.smu.edu/sigmod2000/sigmodcfp.html

The SIGMOD conference will be a forum for database researchers, practitioners, developers and users to explore new ideas and exchange tools, techniques and experiences. W e invite the submission of original research contributions, as well as proposals for panels, tutorials, demonstrations and industrial presentations.

Important Dates: October 11 Paper abstracts due.

October 18 Full and industrial papers as well as panels, tutorials, demos due.

January 24 Notification about acceptance/rejection.

March 1 Camera-ready due.

Organizing Committee and Chairs:

SIGMOD General Chair: Margaret H. Dunham, Southern Methodist Univ.

SIGMOD PC Chair: Jeff Naughton, Univ. of W iscosin

SIGMOD Demo Chair: Jiawei Han, Simon Fraser Univ.

SIGMOD Industrial Chair: Philip Bernstein, Microsoft

SIGMOD Tutorial Chair: Svein-Olaf Hvasshovd,

SIGMOD Panels Chair: Alon Levy, Univ. ofW ashington

SIGMOD Proceedings Chair: W eidong Chen, Southern Methodist Univ.

SIGMOD Electronic Proceedings Chair: Nick Koudas, AT&T Research.

PODS General Chair: Victor Vianu, U.C. San Diego

PODS PC Chair: Georg Gottlob, Vienna Univ. of Technology

PODS Proceedings Chair: Bertram Ludaescher, San Diego Supercomputer Center

Asian Coordinator: Masaru Kitsuregawa, Univ. of Tokyo

European Coordinator: Christian S. Jensen, Alborg Univ.

South American Coordinator: Claudia Bauzer Medeiros, State Univ. of Campinas

Registration Chair: AlexDelis, Polytechnic Univ.

Publicity Chair: Mario A. Nascimento, State Univ. of Campinas

Treasurer: Vijay Kumar, Univ. of Missouri-Kansas City

Local Arrangements Chair: Leonidas Fegaras, Univ. of Texas at Arlington

Corporate Sponsors Chair: Sharma Chakravarthy, Univ. of Florida

1 Refer to the home page for up-to-date information, such as submission details.